Author's response to reviews

**Title:** Gene expression variation to predict 10-year survival in lymph-node-negative breast cancer

**Authors:**

- Elin Karlsson (elin.karlsson@oncology.gu.se)
- Ulla Delle (ulla.delle@oncology.gu.se)
- Anna Danielsson (anna.danielsson@oncology.gu.se)
- Björn Olsson (bjorn.olsson@his.se)
- Frida Abel (frida.abel@genomics.gu.se)
- Per Karlsson (per.karlsson@oncology.gu.se)
- Khalil Helou (khalil.helou@oncology.gu.se)

**Version:** 4  **Date:** 22 August 2008

**Author's response to reviews:** see over
Dear BioMed Central Editorial Production Team,

We were very delighted to hear that BMC Cancer will publish our manuscript, “Gene expression variation to predict 10-year survival in lymph-node-negative breast cancer”.

Your instructions to submit the revised manuscript were carefully followed. We hope that the revised manuscript is now acceptable.

Thank you very much.

Sincerely,

Elin Karlsson